Author's response to reviews

Title: A new use of porcine-derived acellular dermal matrix in primary augmentation mammoplasty to minimize implant-related complications and to achieve an internal mastopexy: a case report series

Authors:

Andrew Kornstein MD, FACS (andrewkornstein@gmail.com)

Version: 4 Date: 14 October 2013

Author's response to reviews: see over
October 14, 2013

Prof Michael Kidd, Editor-in-Chief
*Journal of Medical Case Reports*

Dear Prof Kidd,

I am resubmitting the manuscript “A new use of porcine-derived acellular dermal matrix in primary augmentation mammaplasty to minimize implant-related complications and to achieve an internal mastopexy: a case report series.” The manuscript has been updated based on the latest requests from the editors, namely:

1. Please include the study design in your title, i.e. Case report. For example: A presenting with B in C: a case report.

2. Please change the description of the patient’s gender from ‘woman’ to ‘female’.

3. Please include a list of abbreviations used in the manuscript and their meanings after the Conclusions section.

Please don’t hesitate to contact me should you need further information.

Sincerely,

Andrew Kornstein, MD, FACS
1050 Fifth Avenue New York, NY 10028, USA
Business Telephone: (212) 987-1300
Home Telephone: 917-617-1919
Fax: 212-987-1818
E-mail: andrewkornstein@gmail.com